Author’s response to reviews

Title: Association of 4p14 and 6q27 Variation with Graves Disease: a Case-control Study and a Meta-analysis of Available Evidence

Authors:

Xun Chu (chux@chgc.sh.cn)
Fa-Mei Li (lifamei2007@126.com)
Lin LIU (liulin_sd@sina.com)
Li-Nan Pang (690891659@qq.com)
Min SHEN (shenm@chgc.sh.cn)
Hong-Wen LU (13563612986@163.com)
Xiao-Hong ZHANG (zhangxh71129@163.com)
Zhen-ju Song (zhenjusong@yahoo.com)

Version: 4 Date: 12 Apr 2017

Author’s response to reviews:

April 12, 2017

Dear Dr. Pasini,

Thank you very much for your kind support. Now we are sending you a clean version of our manuscript (MGTC-D-16-00094R4).

As requested, we separated the Acknowledgements and Funding statements. If more information is needed about the manuscript, please don’t hesitate to contact us.

Thanks and best wishes,

Xun CHU, MD & Ph.D.

Department of Genetics, Shanghai-Ministry of Science and Technology (MOST) Key Laboratory of Health and Disease Genomics, Chinese National Human Genome Center at Shanghai